Author's response to reviews

Title: Cross-Sectional Longitudinal Study of the Academic Half-Day Format in a Hematology-Oncology Fellowship Training Program.

Authors:

Ahmed A Eid (aeid@mdanderson.org)
Peggy Hsieh (Peggy.H.Huang@uth.tmc.edu)

Version: 3
Date: 16 February 2015

Author's response to reviews:

Remove the figure at the end of the main manuscript (line 344) as it does not belong there and it has an error (3rd year fellows average score in 2012 was 611 and not 311)